Author's response to reviews

Title: Early prediction of survival after open surgical repair of ruptured abdominal aortic aneurysms

Authors:

Felix Krenzien (felix.krenzien@medizin.uni-leipzig.de)
Ivan Matia (ivan.matia@medizin.uni-leipzig.de)
Georg Wiltberger (georg.wiltberger@medizin.uni-leipzig.de)
Hans-Michael Hau (hansmichael.hau@medizin.uni-leipzig.de)
Moritz Schmelzle (moritz.schmelzle@medizin.uni-leipzig.de)
Sven Jonas (svnins@gmail.com)
Udo X Kaisers (udo.kaisers@medizin.uni-leipzig.de)
Peter T Fellmer (peter.fellmer@medizin.uni-leipzig.de)

Version: 2

Date: 22 August 2014

Author's response to reviews:

Dear Dr. Carlo Banfi,

We are very grateful to the editorial team and the reviewers for their insightful and helpful comments on our manuscript MS: 1662475542124919 entitled “Early prediction of survival after open surgical repair of ruptured abdominal aortic aneurysms”. We appreciated that the reviewers recognized the importance of this work. In this revision, we made every effort to comprehensively and thoroughly address all concerns, and believe that the present manuscript benefited from the reviewers’ insights and as a result is considerably improved. The revision fully addresses all of the specific concerns.

We have enclosed our statements to the reviewers’ along with the revised manuscript. We hope that you will accept our article for publication. Thank you very much for giving us the opportunity to address the reviewers’ comments and to submit a revised manuscript.

Yours sincerely,

Felix Krenzien